Aalborg Universitet



An Experimental Framework for 5G Wireless System Integration into Industry 4.0 Applications

Rodriguez, Ignacio; Mogensen, Rasmus Suhr; Fink, Andreas Engelsen; Raunholt, Taus Mortensen; Markussen, Søren; Christensen, Per Hartmann; Berardinelli, Gilberto; E. Mogensen, Preben; Schou, Casper; Madsen, Ole Published in: Energies

DOI (link to publication from Publisher): 10.3390/en14154444

Creative Commons License CC BY 4.0

Publication date: 2021

Document Version Publisher's PDF, also known as Version of record

Link to publication from Aalborg University

Citation for published version (APA):

Rodriguez, I., Mogensen, R. S., Fink, A. E., Raunholt, T. M., Markussen, S., Christensen, P. H., Berardinelli, G., E. Mogensen, P., Schou, C., & Madsen, O. (2021). An Experimental Framework for 5G Wireless System Integration into Industry 4.0 Applications. *Energies*, *14*(15), Article 4444. https://doi.org/10.3390/en14154444

General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
You may not further distribute the material or use it for any profit-making activity or commercial gain
You may freely distribute the URL identifying the publication in the public portal -





Article An Experimental Framework for 5G Wireless System Integration into Industry 4.0 Applications

Ignacio Rodriguez ^{1,*}, Rasmus Suhr Mogensen ¹, Andreas Fink ¹, Taus Raunholt ¹, Søren Markussen ¹, Per Hartmann Christensen ¹, Gilberto Berardinelli ¹, Preben Mogensen ¹, Casper Schou ², and Ole Madsen ²

- ¹ Department of Electronic Systems, Aalborg University, 9220 Aalborg Øst, Denmark; rsm@es.aau.dk (R.S.M.); aef16@student.aau.dk (A.F.); traunh16@student.aau.dk (T.R.); smarku16@student.aau.dk (S.M.); phc@es.aau.dk (P.H.C.); gb@es.aau.dk (G.B.); pm@es.aau.dk (P.M.)
- ² Department of Materials and Production, Aalborg University, 9220 Aalborg Øst, Denmark; cs@mp.aau.dk (C.S.); om@mp.aau.dk (O.M.)
- * Correspondence: irl@es.aau.dk; Tel.: +45-9940-7394

Abstract: The fourth industrial revolution, or Industry 4.0 (I4.0), makes use of wireless technologies together with other industrial Internet-of-Things (IIoT) technologies, cyber–physical systems (CPS), and edge computing to enable the optimization and the faster re-configuration of industrial production processes. As I4.0 deployments are ramping up, the practical integration of 5G wireless systems with existing industrial applications is being explored in both Industry and Academia, in order to find optimized strategies and to develop guidelines oriented towards ensuring the success of the industrial wireless digitalization process. This paper explores the challenges arisen from such integration between industrial systems and 5G wireless, and presents a framework applicable to achieve a structured and successful integration. The paper aims at describing the different aspects of the framework such as the application operational flow and its associated tools, developed based on analytical and experimental applied research methodologies. The applicability of the framework is illustrated by addressing the integration of 5G technology into a specific industrial use case: the control of autonomous mobile robots. The results indicate that 5G technology can be used for reliable fleet management control of autonomous mobile robots in industrial scenarios, and that 5G can support the migration of the on-board path planning intelligence to the edge-cloud.

Keywords: 5G; Industry 4.0; system integration; prototypes; manufacturing systems; IIoT; autonomous mobile robots

1. Introduction

Seamless system integration is a key aspect of the fourth industrial revolution (Industry 4.0) [1]. In this respect, 5G, one of the most advanced connectivity options, designed to achieve low latency, high reliability, flexibility, and security [2], is expected to be integrated with different cyber–physical systems (CPS), other industrial Internet-of-Things (IIoT) technologies, and cloud computing. This will bring smart factories and other industrial production environments to the next level in terms of optimization of the production processes and flexibility and re-configuration of the industrial manufacturing systems [3]. Common for most of these concepts is that they require extensive knowledge of information technologies (IT) and operational technologies (OT). Furthermore, as such, they should be viewed as a multi-disciplinary venture, making the transition to I4.0 for a particular company, site or factory not an easy task. To assist in this transition, multiple frameworks/methodologies have been proposed, mainly with the aim of defining roadmaps based on the digitalization and technological readiness levels [4–7].

These frameworks aim mainly at helping industrial manufacturing companies in identifying their levels of digital maturity, as well as guiding their digitalization efforts with



Citation: Rodriguez, I.; Mogensen, R.S.; Fink, A.; Raunholt, T; Markussen, S.; Christensen, P.H.; Berardinelli, G.; Mogensen, P.; Schou, C.; Madsen, O. An Experimental Framework for 5G Wireless System Integration into Industry 4.0 Applications. *Energies* 2021, 14, 4444. https://doi.org/ 10.3390/en14154444

Academic Editor: Gwanggil Jeon

Received: 6 June 2021 Accepted: 20 July 2021 Published: 23 July 2021

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2021 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the addition of reflective steps throughout the entire process. In general, these frameworks emphasize that reliable wireless connectivity is the key enabler for system interconnection, allowing for tying all the factory assets together in a seamless way, and facilitating the deployment of advanced IT concepts such as big data and artificial intelligence (AI) but also OT concepts such as autonomous mobile robots or matrix production [7]. However, while some of the frameworks include considerations for the underlying communication-specific aspects, mainly in relation to the industrial application requirements, they do not go into details with the integration and performance that specific wireless technologies may have when applied to specific applications. The wireless landscape in industrial scenarios is comprised by multiple different technologies with very different capabilities, with Wi-Fi being today the de facto connectivity option [8]. However, the performance and reliability of Wi-Fi varies largely depending on the specific environment and application scenario due to its operation over unlicensed frequency bands, where the spectrum is shared with other networks. As in such case, it is necessary to "compete" to access the medium for transmission, high reliability and quality-of-service levels might not be fulfilled under certain conditions [9]. As 5G was designed to provide better performance and higher levels of reliability and, since the first commercial 5G networks for industrial use are being deployed [3], it is now time to ensure that the full potential of 5G is properly evaluated and demonstrated within the context of industrial use [10]. However, we are witnessing a lack of proper use case analyses, performance evaluations and development of guidelines addressing the integration of wireless technologies within the different I4.0 applications but also within the overall digitalization processes.

From a 5G perspective, the envisioned flexibility targeted by the I4.0 concept will require a proper assessment of the integration of the networking elements with the individual applications or use cases [11]. Thus, the objective of this paper is to propose an experimental framework for 5G wireless system integration into Industry 4.0 applications. This framework defines an industrial automation operational flow structured in different steps aiming at demonstrating, evaluating and optimizing the feasibility of operating specific industrial use cases over 5G in realistic factory conditions. The framework considers specific needs from factories or industrial entities, industrial-grade hardware, specific communication requirements and protocols, to evaluate the feasibility of operation of selected use cases over 5G and to benchmark it with alternative wireless technologies. In order to achieve that, the experimental framework is complemented by a specific industrial research lab environment and a number of 5G hardware and software prototyping tools. The specific selections of experimental methodology, available in-lab technologies, and even the specific hardware and software prototyping implementations have been partly impacted by generalization of learnings and inputs from our extensive conversations with multiple entities of the Danish manufacturing industry.

The rest of the article is organized as follows: Section 2 discusses a number of framework design considerations based on the learnings from our conversation with the Danish Industry. Section 3 describes the proposed experimental framework, including the operational flow, research lab facilities and hardware and software prototyping implementations. Section 4 illustrates the applicability of the experimental framework for a specific industrial application: autonomous mobile robots (AMRs). Section 5 provides a discussion on the potential considerations for future versions of the framework. Finally, Section 6 concludes the paper.

2. Learnings from Industry and Related Framework Design Considerations

In order to guarantee that our proposed 5G system integration framework is aligned with the needs of industry, the main learnings from four years of numerous conversations and visits to different Danish factories are analyzed and considered directly into the design:

• The manufacturing industry, and especially the small and medium enterprises, has little or no experience with wireless communications. In general, some experience with Wi-Fi (non-optimized in most cases) was observed, but not with 4G, 5G, and the

other wireless technologies. This is mainly due to the fact that, until recently, products and business models for their usage in factory were practically nonexistent. Thus, an initial learning process should be expected when introducing these technologies in manufacturing environments.

- Design consideration: a common language and understanding needs to be established. Each factory and operational teams are different from others so, ideally, this should be done on an individual factory and use case basis. Benchmarking the performance of the wireless solution, not only over 5G, but also over other technologies such as Wi-Fi should help in improving the understanding of the different wireless operational possibilities. In this case, disseminating performance test results to manufacturing experts is an important aspect to consider.
- Deploying a wireless system is typically not considered in the business models, and is generally perceived as a cost and not as benefit. Moreover, due to the lack of experience with wireless communications, in most cases, a large degree of skepticism about the reliability, capabilities and potential of these technologies was detected.
 - Design consideration: In order to build trust and convince the manufacturing industry about the suitability and the potential benefits of wireless applied to production, live demonstrations of wireless-operated industrial use cases are encouraged. This could take form of trials in factories or demos in lab environments, where wireless-integrated production concepts are shown to manufacturing experts.
- In general, current wired industrial factory control networks are not optimized for direct integration with wireless technologies. Integration gaps range from a nonoptimal topology, to the use of non-IP traffic or transmission modes, such as broadcast, that may not be directly enabled/supported by wireless technologies. To fully exploit the benefits of wireless applied to industrial production, a re-architecture of such control networks will be an essential step.
 - Design consideration: hardware and software prototypes should be flexible enough to cope with modern (IP-based) and old (non-IP-based) communication protocols. Furthermore, the prototypes should be flexible enough to support different control architectures and modes of operation such as infrastructure mode (where end-devices communicate to a centralized entity or controller) and device-to-device communication (where the end-devices can communicate directly among themselves).
- There is a huge hype among the manufacturing industry about cloud control and cloud monitoring, which lead to a noticeable tendency of installing such systems within their production equipment without a proper performance impact assessment, as high capacity communication links available from each device to the cloud are taken for granted. However, this might not always be true, especially when operating over wireless, which could lead to some operational problems.
 - Design consideration: a proper data traffic analysis should be done before proceeding with the integration of an application with wireless in order to identify all potential communication flows in a given system. Further, a performance evaluation should be done after the wireless solution is deployed, to ensure operational correctness.
- Legacy is extremely important. Not all companies will have the chance to invest in the most advanced solutions, but still introducing a few wireless components for specific communication needs might result in a considerable gain for them.
 - Design consideration: hardware and software prototypes should be able to transport over wireless legacy communication protocols that are conceived for industrial wired setups.

3. Experimental Framework for 5G Integration

The proposed experimental framework is aimed at providing services to industries, and is expected to be technology-unbiased. In this respect, the main actors of the framework can be, in a first moment, academic institutions interested in cross-disciplinary research on wireless and manufacturing. Initiating a dialogue with the manufacturing sector, and bringing wireless solution to factories, can indeed be mutually beneficial. The manufacturing sector can obtain an unbiased view on the wireless technologies and installation types that would better suit their specific needs. Besides, the manufacturing sector can enrich their vision for a wireless factory by leveraging the knowledge of the scientific trends brought by the academic institutions. Our dialogue with Danish factories has revealed that such unbiased view on wireless technologies is of great value for them. On the other hand, academic institutions can acquire extra knowledge on real factories setups, data traffic and production needs, besides identifying shortcomings of current wireless solutions or manufacturing concepts in addressing the factory demands. This can pave the way of future research activities on both wireless and manufacturing domains, and their integration. In the long term, we foresee opportunities where new business entities can act as middlemen between the manufacturing sector and the wireless sector (vendors and operators) for providing adequate recommendations on solutions and installations which are not tight to specific technologies or equipment/service providers and, thus, this framework can be of relevance for them.

The proposed framework is composed of: an operational flow that describes the overall sequential actions to be taken to achieve a successful operation of a given I4.0 application over 5G, an industrial research lab environment where the experimentation will take place, and the 5G hardware and software prototypes tools that are used to integrate the 5G system and the machinery associated with the selected industrial use case.

3.1. Industrial Automation Operational Flow

As illustrated in Figure 1, the proposed experimental framework is based on a sequential operational flow structured in six steps. Each of the steps has a different objective and makes different use of the lab environment and prototyping tools that are described in Sections 3.2 and 3.3, respectively.



Figure 1. Flow diagram illustrating the different steps of the proposed industrial automation operational flow.

- STEP 1. Understanding the needs of the factory and its current level of digitalization: the factory is visited in order to get a good overview of overall operations and digitalization status. It is important, that during a first visit, common ground and common languages are established. Conversations are typically initiated with management, digitalization, research and innovation responsibles to learn about their views on 5G, the specific applications that they envision to run in their factories in the future, but also about those current applications that they would like to operate over wireless for improving the current production systems. Conversations are also held with IT responsibilities in order to understand the current wired and Wi-Fi network architectures, office and production network splits and network performance. With this information, we can feed the following steps of the operational flow with realistic information about the specific operational conditions required.
- STEP 2. Understanding the exact communication requirements of the wired setup (data traffic analysis): in this step, that also takes place at the factory, a specific targeted I4.0 application is selected and the production area is visited—ideally, together with someone from both the IT and manufacturing departments, in order to have an overview of the exact machinery setting and its control network architecture. Further, a data traffic analysis is performed over relevant interfaces, candidates to be potentially operated over 5G, to understand and characterize the exact communication flows, protocols used, and the volume of data in different parts of the industrial system. This step is essential, as similar applications in different factories might implement very different control communication systems, and thus the more detailed the information, the better. It is important to check with the IT or production engineers about the control-loop latencies implemented in their systems, as well as survival times (sometimes called control failure times). These two parameters, together with the control architectures and the traffic statistics (throughput, packet sizes and inter-packet times) are essential information for evaluating whether the chosen application would be suitable for operation over 5G.
- STEP 3. Selection of the appropriate wireless technology and dimensioning of the solution: based on the information collected in the previous step, we can analyze the suitability of 5G or other wireless technologies for coping with the targeted application and guide the factory about the most appropriate one. These could be done by simple "pen and paper" exercises based on requirements and wireless capabilities or via simulation/emulation. In general, the decision will be made according to capacity and control-loop latency requirements. High throughput time-critical applications could only be supported by 5G, while those applications that are delay-tolerant could be supported by 5G or other alternative technologies such as 4G or Wi-Fi (under very specific and optimized operational conditions) [9]. It should be noted that some of the most demanding applications might still not be supported by 5G, e.g., applications requiring sub-ms latencies, and in that case, legacy wired setups should still be used.
- STEP 4. Deployment: once the wireless technology has been selected, i.e., 5G, the practical system integration begins. One or more of the relevant interfaces identified in STEP 2 are selected for integration of a 5G interface into them. This process is normally not straight forward as the IT integration itself, and the debugging of connectivity and 5G-integrated interface routing schemes, turns typically into a quite time-consuming task. Ideally, at the end of this step, the result is a fully functional 5G-integrated I4.0 application, both in terms of hardware and software, ready to be tested in operational conditions.
- STEP 5. Performance analysis: the fully functional 5G-integrated industrial application is thoroughly tested. In this step, typically, reliability tests are executed by monitoring the correct operation of the industrial system during a long period

6 of 23

of time (could be from several hours to several days). Performance tests are also done, looking at the communication efficiency in terms of, for example, control-loop latency performance. Other parameters such as throughput or time between failures could be monitored, ensuring that the outcome of the testing is well aligned with the methodology defined in [12], guaranteeing that a fair comparison with future results reported by Industry for similar use cases will possible. Further, the industrial production efficiency could be evaluated, by mapping the empirical wireless performance to the operation of the underlying industrial process [13]. Scalability tests could also be performed, in case they are relevant for the targeted use case. Ideally, similar tests should be done over other suitable wireless technologies (i.e., 4G, Wi-Fi), in order to benchmark their performance with the 5G one and estimate the potential gains achieved by operating the application over 5G.

STEP 6. Optimization: this final step deals with the potential enhancements in those areas of the system identified based on the analysis of the outcome of the different performance tests. Depending on the results, optimizations could take place in different domains: at the machinery side, in case it is detected that an optimized performance would be feasible by a slight re-architecture of the industrial control network or by a slight tuning of parameters in the control communication protocols; at the 5G network side, by optimizing the radio interface for the specific targeted application by tuning of 5G configuration parameters; or at the 5G-application integration point, in case it is identified that changes in the integration hardware or software are needed.

Once STEP 6 in the applied operational flow is completed for a given application, this should be ready to integrate and deploy with the operational production system within the factory. Then, it is possible to go back to STEP 1 and re-initiate the flow after re-evaluating what the new needs of the factory would be after the successful completion of the previous iteration. The introduction of a wireless link may affect the entire networking ecosystem in a factory (eventually unleashing new opportunities for running wirelessly other applications involved in the same control network infrastructure), and therefore the integration of such other application and their potential conversion to wireless should be considered in the next iteration. This converts our operational flow into a circular loop, and makes the proposed experimental operational framework compatible for direct integration with those digitalization methodologies such as the AAU 360 DMA [6], briefly addressed in Section 1. Our integration methodology is flexible enough to be applied to any application at any level digitalization level. Furthermore, despite the fact that in this paper we illustrate and apply it mainly within a 5G context, our framework is technology-neutral. We believe that multiple wireless technologies will co-exist within the industrial ecosystem, and this makes our integration framework unbiased as compared to, for example, those that specific network providers interested in selling a given technology might suggest.

3.2. Industrial Research Lab Facilities

Ideally, all steps in the proposed operational flow should take place at the operational factory. However, this is not always a possibility as one may need to interrupt the manufacturing process for experimentation, which might have a negative impact on the productivity. In some cases, factories might even not have a 5G network deployed at the time when they begin to get interested in the potentials of 5G and how it could be applied for their manufacturing use cases. Therefore, having an industrial research lab with 5G capabilities where STEP 3 and beyond of our proposed flow could be executed is of paramount importance. In some specific cases, even STEPS 1 and 2 could take place in the lab, as sometimes it might be possible that representatives from the factory will visit the lab, tell about their digitalization plans and bring some of the pieces of their manufacturing system that they would like to integrate with 5G, e.g., robotic arms, programmable logical controllers (PLCs), AMRs, sensors and actuators, etc. In those cases where 5G is already available in the factory, the practical 5G integration can take place at either place. For example, for the practical work done in STEP 4, initial efforts can be done at the research lab and, once the system is tested to be stable, the integrated version can be deployed in the factory. This will reduce significantly the impact on the manufacturing process as there would be no need for the factory to interrupt the production for experimentation.

The industrial research lab used in connection with the proposed framework is the "Aalborg University 5G Smart Production Lab", which is a small-scale industrial factory environment composed of two halls, equipped with a wide range of operational industrialgrade manufacturing and production equipment including production lines, welding machines, robotic arms, etc. The key aspect of this research lab is that it is further equipped with a selection of the most advanced wireless technologies from multiple operators and vendors, which are fully available for our integration and testing efforts. Overview pictures of the two industrial halls of the research lab and the different wireless network deployments are shown in Figure 2.



Figure 2. Overview of the Aalborg University 5G Smart Production Lab, including details on the two industrial halls and the different operational wireless network deployments.

In particular, the industrial research lab is equipped with the following state of the art wireless research networks:

- 1× 5G NR private network (5G pNR), 3 cells.
- 1× 5G NR dedicated operator network slice (5G dNR), 3 cells.
- 2× 4G LTE private networks (4G pLTE), 3 cells each.
- 1× 4G LTE dedicated operator network slice (4G dLTE), 3 cells.
- 2× Wi-Fi 6 networks, 3 cells each.
- 1× industrial wireless LAN network (IWLAN), 3 cells.
- 1× LoRaWAN network, 1 gateway.
- $1 \times$ ultra-wideband radio positioning system (UWB), 16 anchors.

Having such a lab with such a selection of industrial equipment and wireless networks is an advantage for us when implementing the integration framework, as it offers a high degree of flexibility and possibilities with respect to choosing where a given step of the operational framework flow should take place. Specifically in STEPS 5, the lab offers a controlled test environment that allows to follow the guidelines from [12] for performance testing of I4.0 use cases over 5G in real-world conditions. Further, the performance achieved over 5G could be benchmarked with that achieved over 4G or Wi-Fi, for example, providing additional input into the suitability of the different wireless technologies for operating a given use case. Similarly for STEP 6, the lab network controlled environments will

facilitate performing some optimizations before moving the use case to production in the factory. Further, our research lab can be used an exhibition room where the operational 5G-integrated prototypes can be showcased to Industry, helping in generating trust on wireless when applied to manufacturing.

3.3. 5G Hardware and Software Prototyping Tools

In order to execute our work throughout the different phases of the experimental operational flow, we developed a number of customized hardware and software prototyping tool devices:

- Network traffic sniffer (NTS): intended to log all data passing through an ethernet interface and extract relevant network traffic statistics.
- 5G Emulator (5GE): aimed at introducing controlled delays into an ethernet link to imitate the communication performance of a wireless technology, e.g., 5G.
- Wireless multi-access gateway (WMAGW): designed as the main integration element to enable the wireless transport, e.g., over 5G, of data traffic from a given ethernet-based interface.

Although there currently exist commercial devices such as network traffic analyzers [14–16], network emulators [17–19] and 5G industrial devices [20–22], with somehow similar purposes and characteristics to the ones custom-developed by us, this was not the case at the time when our 5G integration experimental activities were initiated 4 years back. As it will be explained later for each of the individual prototypes, our custom designs offer us some benefits and flexibility that would not be possible to obtain by using commercial devices.

Our reference designs consider ethernet interfaces for integration to the industrial machinery, as most of the industrial applications that will make use of 5G are currently operating over ethernet cables [23,24]. However, it should be noted, that the design could be slightly adjusted to be integrated with other interfaces such as USB or fieldbuses.

As illustrated in Figure 3, the NTS and the 5GE have a very similar functional structure. Both devices shared an identical hardware (HW) configuration with two bridged ethernet ports that operate by forwarding the data input at one end to the other end. Their software (SW) implementation is, however, different. The NTS (Figure 3a) works as a "man-in-the-middle attack" [25] by executing a SW that logs relevant information from all the L2 and above data traffic packets passing through the bridge interface. Only headers are recorded. No packet inspection is done in order to preserve the confidentiality of the potential business-critical data contained in the payload of the control data messages flowing across the logged ethernet interface. The main use of the NTS box within the operational flow is done in STEP 2, when the data traffic analysis of the selected industrial application is performed. Compared with existing commercial network traffic analyzers, our NTS has similar form factor and functionality to the one developed by ProfiTAP [14]. Other network traffic analyzer solutions such as the ones from CISCO [15] or Solarwinds [16] are much more enterprise-oriented and offer extensive analysis suites, which makes perfect sense for live enterprise network monitoring but would be an overkill for our traffic characterization purpose. The main advantage of building our own analyzer is that the developed SW can be targeted to extract only relevant parameters of our interest such as number of flows, protocols, packets sizes, or packet inter-arrival times, without logging any payload with factory-critical information. This custom and controlled implementation plays in our favor as some of factories would not allow to connect commercial tools that logs and analyzes all their traffic uncontrollably.

The 5GE (Figure 3b) implements a different SW that introduces specific delays to the bridged communication link. The packets received by the 5GE are restrained for a pre-configured delay which is obtained from the uniform sampling of a specific delay distribution pre-loaded from a configuration file. The 5GE can be configured to emulate the delay performance of different wireless technologies by simply loading specific files containing distributions of empirical delay samples for a given technology, e.g., 5G. Such

files are generated by storing empirical samples from real-world lab traces such as the private 5G Rel. 15 ones obtained in [9], or from simulations for those technologies that are not available in the lab, e.g., 5G ultra reliable and low latency communication (URLLC). The interval at which the device adjusts the delay values is fully configurable, although it has not yet been possible to adjust it in a per-packet basis, due to HW limitations. The current implementation adds similar delay in both communication directions, although it should be possible to adjust the SW implementation to introduce different delays for different directions. The 5GE can be used mainly in connection with STEP 3 in the operational flow to estimate the performance of the selected use case over a given wireless technology and validate the suitability of the selection. The 5GE constitutes a simplified intermediate solution between a 5G radio channel network emulator like the one from Keysight [17] and an ethernet network emulator such as the ones from Spirent [18] or iTrinegy [19]. The main benefit from the 5GE as compared to those commercial devices is that we can emulate 5G performance over ethernet-based links based on empirical traces obtained by ourselves in realistic and controlled operational conditions, without having to rely on pre-configured delay and jitter settings or standard channel models. To date, our 5GE implementation only allows for emulation of a single 5G link, while commercial tools are capable of emulating larger networks consisting of multiple links.



Figure 3. (a) Functional block diagram of the NTS. (b) Functional block diagram of the 5GE. (c) Picture of the NTS/5GE (v2) device deployed within a production line module. Under such exact same deployment configuration, the box can be used to either logging the data traffic to and from a PLC, or to introduce communication delays to emulate the connectivity of the PLC over 5G wireless.

Both devices are "plug & play" and their operational procedures are very similar. In order to plug the NTS or the 5GE into industrial machinery as shown in Figure 3c, the target ethernet cable should be disconnected from the machinery (e.g., PLC in the picture) and connected to the one of the ports of the device (e.g., the orange cable that spans from the top to the bottom of the picture). Then, the other port is connected back to the machinery with the original PLC Ethernet cable (e.g., the green cable in the middle of the picture that connects to the network switch on the left). Once the device is installed, it should be powered on and it will automatically begin its operation. As a practical recommendation, the NTS and 5GE should be used over the machinery for a time long enough to capture at least one full production cycle in the system as this would ensure that the data traffic analysis or the 5G emulation captures all possible communication states.

The functional architecture of the WMAGW is displayed in Figure 4a. As detailed in the diagram, the WMAGW enables two different modes of connectivity: IP-based mode or L2-tunnel mode. The use or one or the other mode depends on the characteristics of the

selected industrial application and its associated communication requirements, and the capabilities of the machinery that the WMAGW device will be plug into. By using the WMAGW in L2-tunnel model, we ensure that all traffic at all layers running over the ethernet interface of the production machinery is transported over wireless, e.g., 5G. This is done by encapsulating the incoming ethernet frames into UDP packets, which are later routed over the wireless port interface of the source WMAGW towards a destination WMAGW. The data sent over 5G is received at the destination WMAGW via its wireless interface, where the UDP packets are decapsulated into ethernet frames and sent over the ethernet port of the device into the ethernet interface of the production machinery. If all data control traffic in the selected application is IP-based and there are no L2 control mechanisms implemented in the system (i.e., broadcast), the IP-based mode can be used instead. This mode simplifies the integration setup and the use of the WMAGW as the SW is more computationally efficient than for the L2-tunnel mode, since we are simply bridging the ethernet and wireless interface of the gateways and establishing routing paths, but its application is not always possible especially when addressing legacy production control systems.



Figure 4. (a) Functional block diagram of the WMAGWs (operating in device-to-device L2-tunnel mode). Note: for IP-based operation, the alternative path is chosen via SW configuration. Infrastructure mode is supported by using the exam same functional blocks. (b) Picture of the implemented WMAGW prototype (v2) including details on the different HW components. (c) Picture of the implemented WMAGW prototype (v3) including details on the different HW components.

WMAGWs are used from STEP 4 and above of the operational framework flow. When utilizing the WMAGW in device-to-device communication mode, two boxes are needed; one at each of the machinery devices that are to be connected over 5G. In the case of addressing a centralized infrastructure-mode case, e.g., enabling the 5G communication between a production station and a factory controller located in a server room, only one physical WMAGW device is needed and deployed at the machinery side, while at the other end only the SW is deployed to encapsulate/decapsulate the data.

The implemented functional architecture of the different boxes is the same in all different evolution versions (v). However, the specific choice of HW components has evolved throughout our industrial research and experimentation activities and might be different from version to version, leading to different form factors, as illustrated in Figure 4b,c for two different versions of the WMAGW (v2 and v3). As a reference for the different designs, Table 1 summarizes the main hardware and software components implemented in the latest versions of the different devices.

In the specific case of the WMAGW, the modem and antenna configuration might be different depending on the HW version and the intended use of the device. The WMAGW (v3) allows for the installation of up to four modems, and in its reference design (displayed in Figure 4c) builds up two 5G modems and two Wi-Fi six modems. With this configuration, the WMAGW allows for operation over either of the technologies individually, using one or two paths, or over both technologies enabling multi-access multi-connectivity (for increased reliability) [26]. The flexibility in the choice of operational modems and technologies is a key advantage compared to the industrial devices in the market such as the ones from HMS [20], SIEMENS [21] or Robustel [22] which typically implement and support only closed configurations. Further, the state-of-the-art 5G equipment for industrial use is still undergoing its initial commercialization phase and it is not mature enough yet and might not operate fully as expected. Currently, commercial devices might experience some of the same challenges that we typically encounter in our integration research. Initial interfacing with recently deployed and configured 5G networks typically results in connectivity issues or performance instability that requires dedicated debugging and firmware updates and tuning. Thus, being in control of the full device implementation, as it is the case with the WMAGW, is of great advantage. Being in control of the SW implementation also plays in our favor when, for example, 5G connectivity is stable but debugging is needed for the advanced L2 tunneling or multi-connectivity schemes.

Table 1. Summary of main hardware and software components, and other relevant characteristics for the different prototyping devices.

Prototype	Hardware	Software	Other Characteristics
NTS (v2)	Raspberry Pi 4 Model B [27] Broadcom BCM2711, QC@1.5 GHz/4 GB OS: Raspberry Pi OS Lite NIC: Realtec USB3.0/Gbps ethernet adapter	tcpdump 4.99 [28]	HW TAP functionality Network transparent
5GE (v2)	Raspberry Pi 4 Model B [27] Broadcom BCM2711, QC@1.5 GHz/4 GB OS: Raspberry Pi OS Lite NIC: Realtec USB3.0/Gbps ethernet adapter	netem 2.6 [29] Custom delay generation	Multi-technology support Network transparent
WMAGW (v3)	Gateworks Newport GW6404 [30] Cavium OcteonTX8, QC@1.5GHz/2 GB OS: Ubuntu 20.04.2 NIC: Gbps ethernet port 5G modem: QUECTEL RM500Q [31] 5G modem: SIMCOM 8200 [32] 5G modem: SIMCOM 8300G [33] Wi-Fi 6 modem: Intel AX200NGW [34]	Custom multi-access tunnel Custom IP bridge	Auto-network discovery Multi-connectivity support Gbps/USB3.0 support L2 BC/MC support 5G SA/NSA support

Capabilities and Calibration of the 5G Prototyping Tools

A thorough calibration was done for each of the boxes in order to ensure a correct and reliable performance, quantify their capabilities, and understand their potential limitations. In the following, a number of calibration test results are presented, illustrating mainly the processing delay introduced by the devices in operational conditions. The NTS, 5GE and WMAGW were tested by feeding data traffic into the systems under different configurations of packet sizes and packet inter-arrival times (PIAT) and benchmarking the output against the input in terms of latency and throughput. A total of 18 different realizations with different traffic input configurations were tested for each device by sweeping over the combination of 6 different packet sizes (64, 128, 256, 512, 1024 and 1470) bytes (B) and

3 packet inter-arrival times (PIAT) (1, 10 and 100) ms. Calibration test results were collected over a long-enough time of operation to guarantee that we had enough statistics to claim a reliable accuracy up to the 99.9% level, i.e., 10^{-3} level in the below complementary cumulative distribution functions (CCDF) [35]. To achieve enough statistical confidence at such reliability levels, more than 100,000 samples were collected per realization, fulfilling the requirements set by normal approximation of the Binomial distribution [36]. Further, the results illustrate the statistical distribution when considering all samples from all realizations as well as the difference between best/worse realizations to give an overview of the variability of the system.

As the NTS is typically used in connection with operational industrial machinery, it is important to verify that its use does not impact excessively the performance of the underlying industrial control communication processes while the data traffic is being logged. As illustrated in Figure 5, the processing delay introduced by the NTS device in operational conditions is bounded to 0.45 ms, which would not affect significantly the analyzed control traffic as the introduced latency is deterministic (with small jitter <33 μ s). This jitter sets the practical limitation to what the minimum possible PIAT measured in with the NTS is. Based on the calibrated performance, it can be concluded that industrial operational machines, over which our NTS box is used, will continue to operate reliably as long as the survival-time of the underlying control protocols can tolerate a small misalignment when our device starts to function.



Figure 5. CCDF of the processing delay introduced by the NTS device in operational conditions.

In the case of the 5GE, as the device will be used to emulate the performance of a 5G link between two devices, apart from the processing delay of the device itself, the correctness of the generated delay outputs needs to be validated. Figure 6 displays the results of the latency tests for two different configurations: one where no delay was added, so that the processing delay could be determined, and another one where a 1 ms constant offset delay distribution is configured (i.e., values are read from a file containing 100.000 delay samples of equal value of 1 ms), in order to confirm the correct performance at the output over a constant reference. The calibrated processing delay of this device is very similar to the one observed for the NTS. This is, essentially, due to the fact that both devices are built over a similar HW configuration with a slightly different SW implementation due to their different functionalities. When the specific 1 ms delay distribution was applied to the emulator, the output was shifted by 1 ms, as expected, validating the main functionality of the device. A similar deterministic behavior was observed with/without input, although a slightly larger best/worse case variability was observed in the case the specific delay distribution was applied. This is mainly due to the extra SW execution time introduced by the implemented uniform sampling mechanisms at configured delay re-configuration intervals (100 ms in this test case). In any case, this variability is limited (<80 μ s), which ensures that no excessive artificial jitter is introduced, limiting the potential distortion of the device over the input delay distributions, and resulting in a reliable delay emulation.



Figure 6. CCDF of the processing delay introduced by the 5GE device in operational conditions.

Finally, the performance experienced when utilizing the WMAGW devices for transmitting data traffic over the configurable L2-tunnel was examined. The total processing delay introduced by the two devices that set up the ends of the tunnel is considered in Figure 7. In this case, despite of having the combined effect of the processing delay introduced by the encapsulating and decapsulating devices, the delay is bounded by 0.4 ms at the 99.5 percentile ($10^{-2.5}$). This reduced delay as compared to the one in the NTS and 5GE devices is explained mainly by the different HW choice as, as indicated in Table 1, the WMAGW (v3) is implemented over a Gateworks Newport GW6404 [30] single board computer, with higher processing power capabilities than the one of the Raspberry Pi Model 4B [27] over which the NTS (v2) and the 5GE (v2) operate. As a reference, the processing delay performance of the L2-tunnel when implemented with the previous Raspberry Pi-based WMAGW HW version (v2) is also displayed in the figure. In that case, as expected, the overall processing delay was bounded by ~0.75 ms (approximately twice the maximum processing delay of the NTS and 5GE). The good average processing delay performance in the WMAGW (v3) is slightly worse in some cases, where the processing delay can increase from 0.5 to up to 1.5-2 ms. This happens, at most for 0.5%of the packets transmitted over the tunnel (out of 1000 packets, only five will experience a slightly increased processing delay). As further implementation-specific details, it is of interest to report that, currently, approximately 60% of the WMAGW L2-tunnel processing delay comes from the transmission side of the tunnel where VLAN tagging and the encapsulation takes place, while the 40% remaining is due to decapsulation and packet forwarding at the reception side. There is still some room for SW enhancements, processing coordination, and execution optimization, and future versions of the WMAGW will improve further the processing delay to ensure deterministic performance at all reliability levels.



Figure 7. CCDF of the total processing delay experienced in L2-tunneling operational conditions (with two WMAGW devices).

As the WMAGWs are connectivity prototype devices, it was also necessary to make a more detailed analysis of the supported traffic characteristics. A maximum throughput test was performed in this case, trying to push as much data traffic as possible through the L2-tunnel by sending constant-size packets of (64, 128, 512, 1024 and 1470) B with short PIAT intervals ranging from 10 to 100 μ s. Figure 8 displays the results of the maximum throughput test for the different combinations of packet size and PIAT in terms of packet loss. Our WMAGW (v3) devices support reliably the encapsulation/transmission-reception/decapsulation of data packets with sizes of up to 1470 B and a minimum PIAT of 40 μ s. For more demanding traffic patterns with shorter PIAT, packet loss would start to happen due to HW buffering limitations. Therefore, the maximum throughput supported by the WMAGW devices in L2-tunnel mode is limited to 294 Mbit/s.



Figure 8. PER statistics for different combinations of packet size and PIAT when transmitted between two WMAGW in L2-tunneling operational conditions.

It should be emphasized that the reported WMAGW performance applies to the L2-tunnel mode of operation. When the WMAGW is operated in IP-based mode, the overall link processing delays will be shorter as bridging is a much less processing powerdemanding operation than the encapsulation and VLAN tagging operations. Further, in the IP-based operation mode, the maximum supported throughput will not be depending on SW, but mainly on the board and wireless modem HW interfaces. Therefore, in IP-based mode, the WMAGW (v3) can support a throughput of up to 480 Mbit/s.

Based on the calibration test results, we guarantee the compatibility of our prototype solutions with most of the IIoT applications in terms of data traffic [37]. For quick reference, Table 2 summarizes the main operational performance specifications of the different prototypes. Commercial devices are typically implemented on very optimized HW/SW platforms, which guarantees an optimized performance and thus, processing time calibration values are, normally, not reported in their specification data-sheets. With respect to maximum supported throughput, commercial network traffic analyzers and emulators are typically subject to the capabilities of the ethernet input interfaces. With respect to the commercial industrial 5G devices, in IP-mode, the limitation is typically imposed by the capabilities of the selected modems. As in our implementation, the maximum supported throughput is also reduced for commercial devices if some extra SW is executed, i.e., to operate in L2-tunnel mode. As a reference, the HMS 5G device [20] supports maximum 70 Mbit/s when a 5G virtual private network (VPN) connection is established. Our WMAGW prototype supports higher throughput in L2-tunnel mode as no extra link encryption is implemented (we rely fully on the 5G encryption for security), which results in a smaller communication protocol overhead than in the VPN case.

Prototype	Processing Delay	Jitter	Max. Supported Throughput
NTS (v2)	0.45 ms	33 µs	N/A
5GE (v2)	0.47 ms	33 µs	N/A
WMAGW (v3) L2-tunnel WMAGW (v3) IP-mode	<0.4 ms (99.5%) N/A	103 μs N/A	294 Mbit/s 480 Mbit/s

Table 2. Calibrated operational performance specifications of the implemented prototypes.

The described prototype tools have been successfully used in connection with different industrial production systems. The NTS has been used for recording several industrial control data traffic traces for various machines such as production cells, or quality inspection systems or in operational factories [9]. Based on those measurements, we could further validate the correct and reliable operation of the sniffer prototype. Data flow from protocols such as delay-tolerant industrial control protocols like OPC-UA or CIP, but also other more time-critical such as PROFINET, were successfully logged for several hours without detecting any disruption in the operation of the machinery [38]. To date, the WMAGW devices have been used to successfully operate, over 5G (also Wi-Fi and 4G), the control of an operational production line by providing over 5G wireless the connectivity between a factory manufacturing execution system (MES) controller and the different programmable logical controllers (PLC) present inside of each of the stations of the line [9,39,40].

4. Framework Applicability Example: 5G Autonomous Mobile Robots

In this paper, we exemplify the application of the proposed experimental framework on the integration of 5G technology to one of the most important IIoT applications: autonomous mobile robots [41].

4.1. 5G-Connected Autonomous Mobile Robots

Following the steps of the proposed industrial automation research flow, the 5G-integration of AMRs was done as follows:

- STEP 1. An AMR manufacturer would like to start exploring the integration of their products with 5G technology. For mobility reasons, autonomous mobile robots are wireless-native elements. However, as current AMR products operate over Wi-Fi, which presents issues in terms of reliability and scalability when applied to mobile applications [42], optimized alternative are to be explored. AMR vendors are, in general, aware of the techniques for optimizing Wi-Fi performance, however, the products that they sell to their customers will operate over the Wi-Fi networks of the customers, which status remains unknown and might be unoptimized in many cases. Apart from the increased reliability, having 5G-compatible versions of their products would create new business opportunities for AMR vendors.
- STEP 2. In this case, our industrial partner brought some of their AMR products and related control elements such as the fleet manager (FM) to our lab. The control elements and system architecture were open for us so that we could perform a thorough data traffic analysis of its mode of operation. The targeted AMRs are managed from the centralized FM deployed at infrastructure-side (typically in a server room) and accessible via the Wi-Fi network. From the FM, it is possible to send specific missions with automated paths or set waypoints to navigate to for the AMRs. Although, as wireless-native components, AMRs are based on a wireless-wired hybrid architecture, different from that of the standard static production industrial use cases based only on a wired architecture, the framework can still be applied by identifying the relevant communication interfaces that are subject to be integrated into 5G. In this case, thus interface was the standard AMR communication interface (the same that is used currently in the

Wi-Fi version of the product), over which the data traffic flowing was analyzed with the help of the NTS device. We observed that control-loops between the AMR and the FM are executed once per second, and they make use of the TCP protocol with average packets sizes of 4 kB and 100 B, in FM-AMR and AMR-FM communication direction, respectively. The total throughput of the AMR application is, therefore, approximately 32 kbit/s (for a single robot).

- STEP 3. Based on the throughput, packet sizes and inter-packet times values, it was clear that the current AMR implementation is suitable for operation over 5G and its better mobility and reliability support should result in an improved performance of the AMRs in standard mobility operation conditions than when operated over Wi-Fi.
- STEP 4. To integrate the AMR with 5G, a WMAGW (v3) was used. The prototype device was interfaced to the standard communication interface of the AMR by following the 5G-integrated architecture illustrated in Figure 9a. The WMAGW was configured in IP-mode to operate in infrastructure mode to route the control data traffic between the AMR and the FM. In this case, as the architecture of the use case is quite simple, and there are not many communication flows, the IT integration and debugging of the 5G-integrated solution was quite quick. A picture of the deployed 5G-integrated AMR solution is shown in Figure 9b.
- STEP 5. To validate the correct operation of the 5G-integrated application, a performance test was carried out. The test was performed by having the AMR automatically navigating within the lab over the measurement route indicated in Figure 9c. This specific route was chosen to ensure that the AMR was navigating across the multiple 5G cells and guarantee that a proper assessment of the attachment/reattachment to multiple cells and the connection reliability in mobility conditions was done. The performance of the control of the AMR over 5G was compared to the one over different Wi-Fi configurations. The performance tests focused on two main different domains: control-loop latency performance and packet error rate. The results from the tests are displayed in Figure 10. Given the communication requirements of the AMR, with expected control loops every 1 s, all control-loops with latency higher than that will make the AMR stop and enter in a momentary outage. As indicated in the results, no outage was experienced for the 5G-integrated solution, which exhibited a reliable deterministic performance with median control-loop latency of 11 ms and lower than 25 ms at the 99.9 percentile (10^{-3}) . This was, however, not the case for Wi-Fi 5/6, over which control-loop latencies higher than 1 s were experienced during 0.1–0.3% of the time in those cases where no frequency planning was considered (which resembles the typical Wi-Fi deployment situation in operational factories where the AMRs are expected to operate). During these outage occurrences, control-loop latencies close to up to 10 s were observed. These long control-loop latencies will keep the AMR stopped and non-operational for a few seconds every time they happen, which will have an impact of the underlying industrial production process in which AMRs are involved. The performance of the AMR over optimized Wi-Fi 6 and ideal frequency planning (i.e., dedicated spectrum channels per access point) was also observed to fulfill the control-loop requirements for the given use case. However, such performance was less deterministic than the one observed for the 5G-integrated solution. While a median control-loop latency, better than the 5G one, of 5 ms was observed; the value at the 99.9 percentile is increased to over 0.5 s, illustrating the higher reliability of 5G as compared to Wi-Fi. Further, the 5G-integrated solution experienced no packet loss, while in the Wi-Fi case, the PER was 0.4–0.26%.



Figure 9. (a) Architecture of the 5G-integrated AMR solution including the AMR, WMAGW, 5G network infrastructure and AMR fleet manager controller. (b) Picture of the 5G-integrated AMR solution deployed at the lab for the performance testing over the private 5G network and benchmarking with Wi-Fi 6. (c) Lab plan overview including the location of the 5G base stations and Wi-Fi 6 access points, as well as the measurement route followed to test the performance of the AMR in realistic mobility conditions.



Figure 10. CCDF of the AMR-FM control-loop latencies for the different Wi-Fi and 5G network configuration explored in the performance measurement test.

STEP 6. Once it was verified that it is possible to operate AMRs over 5G, the next step would be to look into potential optimizations. Based on the experimental results and the practical observations done along the different operational steps, the potential areas of optimization for the 5G-integrated solution could be: the parametrization of the 5G network configuration to optimize the control-loop latency and reduce the tails of the distribution, or the re-design of the AMR communication schemes to make them more efficient and scalable when operated over wireless in general, or over 5G in particular.

By completion of the operational flow, it was demonstrated the 5G-integration of AMRs. Apart from the commented optimizations, having 5G-connected AMRs opens for new possibilities within the same application ecosystem, such as making use of the 5G capacity and contained latency to migrate some of the intelligence from the AMR to the 5G mobile edge-cloud (MEC).

4.2. 5G Mobile Edge-Cloud Planner for Autonomous Mobile Robots

Following the operational flow, after completion of the first iteration, a new one can be started prior re-evaluation of the new digitalization needs for AMRs. In this second iteration, the 5G-integration flow was applied as follows:

- STEP 1. Once the reliable 5G-control was demonstrated to the AMR manufacturer, they were immediately interested in trying to unleash the full 5G potential by investigating whether other more heavy processes than the FM-based control could be operated over 5G. Moving part of the current on-board intelligence from the AMR to the mobile edge-cloud (MEC) would result in cheaper products and more flexible AMR products as the on-board processing power could be reduced and having a single high-processing power centralized server deployed at infrastructure-side would enable cloud-in-the-loop operation [43] towards all the AMRs, facilitating the exchange of information between AMRs and reducing the burden of having to push SW upgrades to each AMR individually. The specific application selected to be moved to the 5G MEC was the AMR planner. Currently, an AMR navigates by using its on-board planner and does not share any information with other robots. This typically results in suboptimal navigation routes in the case that the AMR navigates into an obstacle and needs to re-plan its route, even if that same obstacle was previously detected and avoided by another robot. This problem can be mitigated by moving the planner functionalities to the edge-cloud where a centralized virtual sharedworld could be built, allowing for the optimization of the navigation for all AMRs at once.
- STEP 2. The AMR vendor provided us with an external planner unit and re-architectured the AMR communication (mainly the I/O connections) to make use of the external device instead of its on-board planner. The new architecture is illustrated in Figure 11a, and can be compared for further reference to the standard AMR architecture in Figure 9a. All internal communication within the AMR happens over ethernet, so the NTS device was used to analyze the data traffic in the I/O-planner communication link. TCP traffic was found with average packet sizes of 128 B and average PIAT of 0.5 ms in both the I/O-planner and planner-I/O communication directions. The overall throughput was 1.3–1.9 Mbit/s (for a single robot), which confirms that, as expected, the I/O-planner commutation is more demanding than AMR-FM one.



Figure 11. (a) Architecture of the 5G-integrated MEC AMR planner solution including the AMR, WMAGW, 5G network infrastructure, and AMR fleet manager controller and planner deployed in edge-cloud configuration. (b) Picture of the 5G-integrated MEC AMR planner solution deployed for 5G emulation at the lab.

STEP 3. Although more demanding than in the AMR-FM case, the I/O-planner link is also suitable for 5G operation, at least based on the throughput, packet sizes and inter-packet times values. However, it was questioned what the impact of the higher 5G delays (compared to the cabled ethernet reference) would be on the performance of the robot. Understanding this is of paramount importance, as the timely and reliable operation of the planner is application-critical, and a excessively long I/O-planner communication delay could result in a sub-optimal navigation performance of the robot. In this case, before proceeding to the deployment step, it was decided to make a 5G-based I/O-planner emulation test to understand the potential impact of 5G delays into the functional operation of the AMR. To do that, the 5GE device was used. As displayed in Figure 11b, the emulator was placed between the I/O of the robot and the external planner device, mirroring the 5G MEC planner configuration illustrated in Figure 11a. A navigation performance and a docking accuracy tests were performed for benchmarking the operation of the reference cabled planner (REF) to the one achieved over 5G. For 5G two different delay distributions were loaded into the 5GE, one with empirical values obtained over 5G Rel. 15 (current 5G commercial version) with an average delay of 4.9 ms, and one with values from a simulation of 5G URLLC [9] with an average delay of 0.75 ms. In the navigation test, a mission was configured to make the AMR navigate over an obstructed route between two selected waypoints within the lab, ensuring that the robot needed to re-plan its path upon detection of the obstacle. The route was covered 40 times in order to ensure that the overall mission times were long enough to observe any potential differences between the reference operation with cabled planner and the 5G-emulated one. For the docking accuracy test, the robot was set to execute a docking maneuver into its charging station from a starting point located at 1 m distance. This test was repeated 15 times in order to obtain an insight into the average experienced accuracy over 5G as compared to the REF. The docking accuracy was evaluated with the assistance of an external optical positioning system with mm-accuracy [44]. The results from the navigation performance tests are presented in Figure 12a, and they indicate that a very small increase in navigation time of 1.5% is expected in the worst case with the 5G MEC planner as compared to its reference operation with the on-board planner. For the



docking accuracy test, the results are shown in Figure 12b, and they illustrate that the use of the 5G MEC planner would result in a slight loss of accuracy of maximum 1–3 mm, which would be negligible for reliable operation of the AMRs.

Figure 12. Results from the MEC-planner performance for the cabled reference case and the different 5G network configurations explored in the emulation tests: (a) navigation performance, and (b) docking performance.

Upon completion of STEP 3, where it was demonstrated by emulation that the AMR would continue to operate in an accurate and reliable manner when having its planner operating in 5G edge-cloud configuration, it would be time to move to the 5G-integration and deployment in STEP 4. However, no further details will be reported for this specific case as the work is still ongoing.

5. Considerations for Future Framework Versions

It has been demonstrated that applying the experimental framework leads to successful integration of 5G into industrial applications. In its current shape, the considered operational flow, prototyping tools and research lab environments, provides enough capabilities for compatibility with 5G Rel 15. However, for future 5G releases, such as those introducing URLLC features and close-to-ms control-loop supports [45], the hardware and software of the prototypes (especially the WMAGW) would need to be adjusted in order to improve, mainly, the processing delays. To achieve this, the HW prototypes will be evolved from the current central processing unit (CPU)-based designs to field-programmable gate array (FPGA) architectures to ensure an optimized processing performance. Using FPGA-based HW elements will require the development of new dedicated SW, as not all the functionalities of the CPU-based SW will be compatible with the new architecture. Such upgrade to high-end performance devices will also ensure that our future HW prototypes will be capable of supporting the integration with time-sensitive networks (TSN) aimed at the control of real-time streams with deterministic delay support [46].

The research lab would also need to be upgraded, in order to include all those 5G network release upgrades. In particular for the lab, 5G base stations with mm-wave FR2 support is seen as an essential evolution step. Under the current 5G lab deployments, operating in sub-6 GHz FR1 spectrum, available capacity might be limited for certain I4.0 applications, especially those demanding high uplink throughput for many devices, e.g., the industrial visual quality inspection based on computer vision [47]. In terms of operational flow, no updates are foreseen in the near future. The operational integration steps are described in a detailed, flexible and technology-unbiased style, which ensures the compatibility with different technologies, industrial use cases, and other reference frameworks.

Security has not been explicitly addressed in our experimental framework, as our main focus is on the 5G integration and connectivity aspects, while security is mainly addressed at higher levels [48]. 5G security in I4.0 application scenarios will depend on various aspects such as the specific flavor of 5G network deployment. A fully private 5G network deployment will have the security advantage of being isolated from the exterior (although

Internet access might exist via a secure firewall), while other 5G deployments for industrial use, such as dedicated network slices, will rely on several external network elements, and as such extra security measures need to be taken [48]. 5G radio communication makes use of encryption and is secure by design, and thus the main security challenges arise mainly from the networking perspective of the integrated end-to-end 5G-operational technology networks [49]. In this respect, our WMAGW will play a role and, thus, should be a secure interface at a similar level of current critical industrial networking infrastructure such as router or switches. Therefore, control access through physical and remote-logical interfaces should be protected by strong credentials. In terms of operation, our WMAGW should be already capable of transmitting ethernet traffic protected by high-layer cryptographic protocols such as DTLS over 5G. However, future developments in terms of industrial networking will be followed in order to ensure that our device continues to perform reliably when operating new secure ethernet protocols such as OPC-UA.

Although not related to the current integration framework itself, the experimentation could lead to identifying and highlighting potential bottlenecks of current 5G networks and, therefore, provide input to wireless, manufacturing and also automation&control engineers and researchers. Wireless researchers could then address such limitations in future radio technologies, e.g., 6G [50], while manufacturing and automation&control researchers could adapt their application and control algorithms to the wireless capabilities. Furthermore, joint manufacturing, communication and control co-design is to be explored.

6. Conclusions

This paper presented an experimental framework for 5G wireless system integration into industrial applications, aimed at providing service to industries, motivated from the lack of digitalization reference models considering in depth wireless performance integration and performance. The presented experimental framework consists on an operational flow that describes the different steps to be applied to move from the understanding of the needs of a factory, to the deployment and optimization of 5G-integrated applications. The framework is complemented by a number of prototyping tools: a network traffic sniffer, a 5G emulator, and a wireless multi-access gateway, which are of a great utility throughout the integration process. The prototyping devices have been carefully designed and calibrated to ensure an optimal behavior and a negligible impact when interconnected with operation industrial machines.

Further, this paper exemplifies the application of the framework on a real industrial use case: the 5G-integration of autonomous mobile robots. Two looped runs of the operational flow are applied: the first focusing on the overall 5G control of the robots, and the second one focusing on the 5G mobile edge-cloud operation of the robot path planner. In both cases, the potential of 5G for operating reliably the use case of autonomous mobile robots was demonstrated. It was also shown how the 5G operation of the robots was superior in terms of control-loop reliability to the one of Wi-Fi 6, which resulted in several robot outages of a few seconds.

The current flow, prototyping tools and lab environment are proven to be effective at current, but the experimental framework will need to continue evolving in order to support the integration of future 5G releases, time-sensitive networks, ultra-low latency control-loop features, or even future more advanced wireless technologies such as 6G.

Author Contributions: Conceptualization, I.R.; methodology, I.R., R.S.M. and P.M.; hardware and software, R.S.M., P.H.C., A.F., T.R. and S.M.; validation, R.S.M., A.F., T.R. and S.M.; investigation, I.R., R.S.M., A.F., T.R., S.M., G.B. and C.S.; writing—original draft preparation, I.R., R.S.M. and A.F.; writing—review and editing, I.R., R.S.M., A.F., G.B. and C.S.; supervision, I.R., P.M., C.S. and O.M.; project administration, I.R.; funding acquisition, P.M. and O.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research was partially funded by Industriens Fond Denmark through the project "Real-Time Industrial IoT with 5G".

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- 1. Aceto, G.; Persico, V.; Pescape, A. A Survey on Information and Communication Technologies for Industry 4.0: State-of-the-Art, Taxonomies, Perspectives, and Challenges. *IEEE Commun. Surv. Tutor.* **2019**, *21*, 3467–3501. [CrossRef]
- 2. Varga, P.; Peto, J.; Franko, A.; Balla, D.; Haja, D.; Janky, F.; Soos, G.; Ficzere, D.; Maliosz, M.; Toka, L. 5G support for Industrial IoT Applications Challenges, Solutions, and Research gaps. *Sensors* **2020**, *20*, 828. [CrossRef] [PubMed]
- 3. O'Connell, E.; Moore, D.; Newe, T. Challenges Associated with Implementing 5G in Manufacturing. *Telecom* 2020, *1*, 48–67. [CrossRef]
- 4. Sanchez, M.; Exposito, E.; Aguilar, J. Industry 4.0: Survey from a system integration perspective. *Int. J. Comput. Integr. Manuf.* **2020**, *33*, 1017–1041. [CrossRef]
- 5. *The Reference Architectural Model Industrie* 4.0 (*RAMI* 4.0), version 1.0; Zentralverband Elektrotechnik- und Elektronikindustrie (ZVEI): Frankfurt am Main, Germany, 2015.
- 6. Colli, M.; Berger, U.; Bockholta, M.; Madsen, O.; Møller, C.; Wæhrens, B.V. A maturity assessment approach for conceiving context-specific roadmaps in the Industry 4.0 era. *Annu. Rev. Control* **2019**, *48*, 165–177. [CrossRef]
- 7. Colli, M.; Madsen, O.; Berger, U.; Møller, C.; Wæhrens, B.W.; Bockholt, M. Contextualizing the outcome of a maturity assessment for Industry 4.0. *IFAC-PapersOnLine* **2018**, *51*, 1347–1352. [CrossRef]
- 8. Seferagic, A.; Famaey, J.; De Poorter, E.; Hoebeke, J. Survey on Wireless Technology Trade-Offs for the Industrial Internet of Things. *Sensors* 2020, *20*, 488. [CrossRef] [PubMed]
- Rodriguez, I.; Mogensen, R.S.; Schjørring, A.; Razzaghpour, M.; Maldonado, R.; Berardinelli, G.; Adeogun, R.; Christensen, P.H.; Mogensen, P.; Madsen, O.; et al. 5G Swarm Production: Advanced Industrial Manufacturing Concepts enabled by Wireless Automation. *IEEE Commun. Mag.* 2021, 59, 48–54. [CrossRef]
- 10. Gidlund, M.; Lennvall, T.; Akerberg, J. Will 5G become yet another wireless technology for industrial automation? In Proceedings of the IEEE International Conference on Industrial Technology (ICIT), Toronto, ON, Canada, 22–25 March 2017; pp. 1319–1324.
- 11. Spinelli, F.; Mancuso, V. Toward Enabled Industrial Verticals in 5G: A Survey on MEC-Based Approaches to Provisioning and Flexibility. *IEEE Commun. Surv. Tutor.* **2021**, *23*, 596–630. [CrossRef]
- 12. 5G Alliance for Connected Industries and Automation (5G-ACIA). White Paper: Performance Testing of 5G Systems for Industrial Automation. Available online: https://5g-acia.org/whitepapers/performance-testing-of-5g-systems-for-industrial-automation-2/ (accessed on 16 July 2021).
- 13. Mogensen, R.S.; Rodriguez, I.; Schou, C.; Mortensen, S.; Sørensen, M.S. Evaluation of the Impact of Wireless Communication in Production via Factory Digital Twins. *Manuf. Lett.* **2021**, *28*, 1–5. [CrossRef]
- 14. ProfiTAP ProfiShark Network TAPs. Available online: https://www.profitap.com/profishark-network-taps/ (accessed on 16 July 2021).
- 15. CISCO Security Packet Analyzer. Available online: https://www.cisco.com/c/en/us/products/security/security-packetanalyzer/index.html (accessed on 16 July 2021).
- 16. Solarwinds Network Traffic Analyzer. Available online: https://www.solarwinds.com/netflow-traffic-analyzer (accessed on 16 July 2021).
- 17. Keysight 5G Network Emulation. Available online: https://www.keysight.com/es/en/cmp/2020/5g-network-emulation-software.html (accessed on 16 July 2021).
- 18. Spirent Network Emulator. Available online: https://www.spirent.com/products/network-impairment-emulation-testing (accessed on 16 July 2021).
- 19. iTrinegy NE-ONE Professional Network Emulator. Available online: https://itrinegy.com/ne-one-professional-range/ (accessed on 16 July 2021).
- HMS Wireless Router 5G NV1000. Available online: https://www.hms-networks.com/docs/librariesprovider6/labs/routeronly.pdf (accessed on 16 July 2021).
- 21. SIEMENS SCALANCE MUM856-1. Available online: https://new.siemens.com/global/en/products/automation/industrial-communication/industrial-5g/5g-router-scalance-mum.html (accessed on 16 July 2021).
- 22. Robustel 5G IoT Router R5020. Available online: https://www.robustel.com/en/product/r5020/ (accessed on 16 July 2021).
- 23. 5G Alliance for Connected Industries and Automation (5G-ACIA). White Paper: Integration of Industrial ethernet Networks with 5G Networks. Available online: https://5g-acia.org/wp-content/uploads/2021/04/5G-ACIA_Integration-of-Industrial-Ethernet-Networks-with-5G-Networks-.pdf (accessed on 16 July 2021).
- 24. Wollschlaeger, M; Sauter, T.; Jasperneite, J. The Future of Industrial Communication: Automation Networks in the Era of the Internet of Things and Industry 4.0. *IEEE Ind. Electron. Mag.* **2017**, *11*, 17–27. [CrossRef]
- 25. Baud, M.; Felser, M. Profinet IO-Device Emulator based on the Man-in-the-middle Attack. In Proceedings of the IEEE Conference on Emerging Technologies and Factory Automation (ETFA), Prague, Czech Republic, 20–22 September 2006.

- Suer, M.T.; Thein, C.; Tchouankem, H.; Wolf, L. Multi-Connectivity as an Enabler for Reliable Low Latency Communications—An Overview. *IEEE Commun. Surv. Tutor.* 2020, 22, 156–169. [CrossRef]
- 27. Raspberry Pi 4 Model B. Available online: https://www.raspberrypi.org/products/raspberry-pi-4-model-b/specifications/ (accessed on 4 June 2021).
- 28. tcpdump. Public Repository. Available online: https://www.tcpdump.org/ (accessed on 4 June 2021).
- 29. netem. Network Emulator. Linux Traffic Control. Available online: https://wiki.linuxfoundation.org/networking/netem (accessed on 4 June 2021).
- Gateworks Newport GW6404. Available online: https://www.gateworks.com/products/industrial-single-board-computers/ octeon-tx-single-board-computers-gateworks-newport/gw6400-single-board-computer/ (accessed on 16 July 2021).
- 31. QUECTEL 5G RM500Q. Available online: https://www.quectel.com/product/5g-rm500q-gl/ (accessed on 4 June 2021).
- 32. SIMCOM 5G SIM8200G. Available online: https://www.simcom.com/product/SIM8200G.html (accessed on 4 June 2021).
- 33. SIMCOM 5G SIM8300G. Available online: https://www.simcom.com/product/SIM8300G_M2.html (accessed on 4 June 2021).
- 34. Intel Wi-Fi 6 AX200. Available online: https://www.intel.com/content/www/us/en/products/docs/wireless/wi-fi-6-ax200 -module-brief.html (accessed on 4 June 2021).
- 35. Herlich, M.; Maier, C. Measuring and Monitoring Reliability of Wireless Networks. *IEEE Commun. Mag.* 2021, 59, 76–81. [CrossRef]
- Brown, L.D.; Cai, T.T.; Dasgupta, A. Confidence intervals for a binomial proportion and asymptotic expansions. *Annu. Stat.* 2002, 30, 160–201. [CrossRef]
- Navarro-Ortiz, J.; Romero-Diaz, P.; Sendra, S.; Ameigeiras, P.; Ramos-Munoz, J.J.; Lopez-Soler, J.M. A survey on 5G usage scenarios and traffic models. *IEEE Commun. Surv. Tutor.* 2020, 22, 905–929. [CrossRef]
- 38. Mogensen, R.S.; Rodriguez, I.; Berardinelli, G.; Pocovi, G.; Kolding, T. Empirical IIoT Data Traffic Analysis and Comparison to 3GPP 5G Models. In Proceedings of the IEEE Vehicular Technology Conference (VCT), Online Conference, 27–30 September 2021.
- Rodriguez, I.; Mogensen, R.S.; Khatib, E.J.; Berardinelli, G.; Mogensen, P.; Madsen, O.; Møller, C. On the Design of a Wireless MES Solution for the Factories of the Future. In Proceedings of the Global IoT Summit (GIoTS), Aarhus, Denmark, 17–21 June 2019.
- 40. Mogensen, R.S.; Rodriguez, I.; Berardinelli, G.; Fink, A.; Marcker, R.; Markussen, S.A.; Raunholt, T.; Kolding, T.E.; Pocovi, G.; Barbera, S. Implementation and Trial Evaluation of a Wireless Manufacturing Execution System for Industry 4.0. In Proceedings of the IEEE Vehicular Technology Conference (VTC), Honolulu, HI, USA, 22–25 September 2019.
- Oyekanlu, E.A.; Smith, A.C.; Thomas, W.P.; Mulroy, G.; Hitesh, D.; Ramsey, M.; Kuhn, D.J.; McGhinnis, J.D.; Buonavita, S.C.; Looper, N.A.; et al. A Review of Recent Advances in Automated Guided Vehicle Technologies: Integration Challenges and Research Areas for 5G-Based Smart Manufacturing Applications. *IEEE Access* 2020, *8*, 202312–202353. [CrossRef]
- Fink, A.; Mogensen, R.S.; Rodriguez, I.; Kolding, T.; Karstensen, A.; Pocovi, G. Empirical Performance Evaluation of Enterprise Wi-Fi for IIoT Applications Requiring Mobility. In Proceedings of the European Wireless (EW), Verona, Italy, 10–12 November 2021.
- Kehl, P.; Lange, D.; Maurer, F.K.; Nemeth, G.; Overbeck, D.; Jung, S.; Konig, N.; Schmitt, R.H. Comparison of 5G Enabled Control Loops for Production. In Proceedings of the IEEE Annual International Symposium on Personal, Indoor and Mobile Radio Communications (PIMRC), Online Conference, 31 August–3 September 2020.
- 44. Raunholt, T.; Rodriguez, I.; Mogensen, P.; Larsen, M. Towards a 5G Mobile Edge Cloud Planner for Autonomous Mobile Robots. In Proceedings of the IEEE Vehicular Technology Conference (VTC), Online Conference, 27–30 September 2021.
- 45. Pocovi, G.; Shariatmadari, H.; Berardinelli, G.; Pedersen, K.; Steiner, J.; Li, Z. Achieving Ultra-Reliable Low-Latency Communications: Challenges and Envisioned System Enhancements. *IEEE Netw.* **2018**, *32*, 8–15. [CrossRef]
- 46. Gundall, M.; Huber, C.; Rost, P.; Halfmann, R.; Schotten, H.D. Integration of 5G with TSN as Prerequisite for a Highly Flexible Future Industrial Automation: Time Synchronization based on IEEE 802.1AS. In Proceedings of the Annual Conference of the IEEE Industrial Electronics Society (IECON), Online Conference, 18–21 October 2020.
- Lu, T.; Fan, Z.; Lei, Y.; Shang, Y.; Wang, C. The Edge Computing Cloud Architecture Based on 5G Network for Industrial Vision Detection. In Proceedings of the IEEE 6th International Conference on Big Data Analytics (ICBDA), Online Conference, 5–8 March 2021.
- 48. 5G Alliance for Connected Industries and Automation (5G-ACIA). White Paper: Security Aspects of 5G for Industrial Networks. Available online: https://5g-acia.org/whitepapers/security-aspects-of-5g-for-industrial-networks/ (accessed on 16 July 2021).
- 49. Tuptuk, N.; Hailes, S. Security of smart manufacturing systems. J. Manuf. Syst. 2018, 47, 93–106. [CrossRef]
- 50. Adeogun, R.; Berardinelli, G.; Mogensen, P.; Rodriguez, I.; Razzaghpour, M. Towards 6G in-X subnetworks with sub-millisecond communication cycles and extreme reliability. *IEEE Access* 2020, *8*, 110172–110188. [CrossRef]